Author's response to reviews

Title: Prevalence of Attention Deficit/Hyperactivity Disorder Among Adults in Obesity Treatment

Authors:

Jules R Altfas (jaltfas@earthlink.net)

Version: 6 Date: 12 Sep 2002

1. Format corrections.

2. Spelling errors corrected.

3. Abstract edited to assure 250 words.